Author’s response to reviews

Title: NEOSCOPE: A randomised Phase II study of induction chemotherapy followed by either oxaliplatin/capecitabine or paclitaxel/carboplatin based chemoradiation as pre-operative regimen for resectable oesophageal adenocarcinoma

Authors:

Somnath Mukherjee (somnath.mukherjee@oncology.ox.ac.uk)
Christopher N Hurt (hurtn@cardiff.ac.uk)
Sarah Gwynne (Sarah.Gwynne@wales.nhs.uk)
Andrew Bateman (A.R.Bateman@soton.ac.uk)
Simon Gollins (simon.gollins@wales.nhs.uk)
Ganesh Radhakrishna (g.radhakrishna@nhs.net)
Maria Hawkins (maria.hawkins@oncology.ox.ac.uk)
Jo Canham (CanhamJ@cardiff.ac.uk)
Wyn Lewis (wyn.lewis4@wales.nhs.uk)
Heike Grabsch (heike.grabsch@mumc.nl)
Ricky Sharma (ricky.sharma@oncology.ox.ac.uk)
Wendy Wade (wendy.wade@wales.nhs.uk)
Rhydian Maggs (Rhydian.Maggs@wales.nhs.uk)
Bethan Tranter (Bethan.Tranter@wales.nhs.uk)
Ashley Roberts (Ashley.Roberts@CardiffandVale.wales.nhs.uk)
David Sebag-Montefiore (d.sebag-montefiore@leeds.ac.uk)
Tim Maughan (tim.maughan@oncology.ox.ac.uk)
Gareth Griffiths (G.O.Griffiths@soton.ac.uk)
Tom Crosby (tom.crosby@wales.nhs.uk)

Version: 2
Date: 20 November 2014

Author’s response to reviews: see over
Re: MS: 1056788232144782

Dear Editor

Thank you for your email dated 20 November. Please find our response to your points in red below.
We have not amended our manuscript.

1. Ethical and Funding Approval Documentation

Before we can proceed with your submission, can you please forward copies of all ethical approval and funding approval for our records. These documents should be sent as email attachments to the following email address, BMCSeriesEditorial@biomedcentral.com. Please DO NOT upload these documents as additional files in the submission system. If your documents are not in English, please could you provide translated versions of the relevant parts. These should be endorsed and signed by a contactable person at the institution. Please also include the original documents. Can you please confirm in your cover letter that you have forwarded the requested documentation to BMCSeriesEditorial@biomedcentral.com.

Confirmed sent to that email address

2. Funding

A study is considered to be externally funded if the authors have been awarded a grant for the study by a major funding body (e.g. governmental funding/ award from a charitable foundation). If a study has not received external funding, then the study protocol will be sent for peer-review with a member of our Editorial Board. If a study has received funding/assistance from a commercial organization, this should be clearly stated in the ‘competing interests’ section of your manuscript, and the study protocol will be sent for peer-review by a member of our Editorial Board. Can you please confirm whether your study protocol has undergone peer-review by the funding body.

Confirmed – reviewed by Cancer Research UK CTAAC Committee (see evidence in email sent to BMCSeriesEditorial@biomedcentral.com).

3. Study status

The protocol must be for a study that is ongoing. An ‘ongoing’ study is defined as one where the investigators are still collecting, or analyzing data. Can you please confirm what stage your study is currently at.

This study is still recruiting

4. Related Articles

Can you please clarify whether any publications containing the results of this study have already been published or submitted to any journal. If so, can you please provide a list of the related articles.
No they have not

Yours faithfully

Chris Hurt